Author's response to reviews

Title: Pre-Invasive And Invasive Disease In Women With A Cytological Diagnosis Of High-Grade Lesion And High Grade Lesion Can Not Exclude Microinvasion

Authors:

Nina S Kuperman (nskuperman@yahoo.com.br)
Fábio B Russomano (fabiorussomano@gmail.com)
Yara F Melo (yarafurtadori@terra.com.br)
Saint Clair S Gomes Jr (scgomes@iff.fiocruz.br)

Version: 2 Date: 7 September 2014

Author's response to reviews:

In the reviewed version I added the email addresses of the authors, a conclusion to the Abstract, a Competing Interest section and an Authors’ Contribution section.